Author’s response to reviews

Title: Association analysis between the tag single nucleotide polymorphisms of DENND1A and the risk of polycystic ovary syndrome in Chinese Han women

Authors:

yanan zhu (zh-yanan@163.com)
Yi-ting Zhang (zhangyiting90@qq.com)
Qin Liu (1398538402@qq.com)
Shan-Mei Shen (shanmeishen@126.com)
Xiang Zou (xiang.y.zou@polyu.edu.hk)
Yun-xia Cao (caoyunxia6@126.com)
Wen-jun Wang (wenjungzcn@163.com)
Long Yi (yilong@nju.edu.cn)
Qian Gao (qian-gao@nju.edu.cn)
Wei-Dong Yang (ywdong2001@sohu.com)
Yong Wang (yongwang@nju.edu.cn)

Version: 1 Date: 17 May 2019

Author’s response to reviews:

Dear editor,

Thank you for the comments from the editor. According to the comments we have carried out the corrections and submitted the revised manuscript. We checked the authors’ information again and we are sure there will be no changes. The manuscript has been reviewed by American Journal Experts for the language editing. Please let us know if there are any problems.

Thank you and best regards.

Yours sincerely,

Yong Wang,
Medical school, Nanjing University